DNB Working Paper

No. 662 / December 2019

Financial knowledge and trust in financial institutions

Carin van der Cruijsen, Jakob de Haan and Ria Roerink

DeNederlandscheBank

EUROSYSTEEM

Financial knowledge and trust in financial institutions

Carin van der Cruijsen, Jakob de Haan and Ria Roerink*

* Views expressed are those of the authors and do not necessarily reflect official positions of De Nederlandsche Bank.

Working Paper No. 662

De Nederlandsche Bank NV P.O. Box 98 1000 AB AMSTERDAM The Netherlands

December 2019

Financial knowledge and trust in financial institutions*

Carin van der Cruijsen^{a*}, Jakob de Haan^{a, b, c} and Ria Roerink^a

^a De Nederlandsche Bank (DNB), The Netherlands ^b University of Groningen, The Netherlands ^c CESifo, Munich, Germany

December, 2019

Abstract

Using fourteen years of data on Dutch consumers' trust in financial institutions, we find that financially literate consumers are more likely to trust banks, insurance companies and pension funds, and the competence and integrity of the managers of these institutions. This holds both for broad-scope and narrow-scope trust. Although trust in respondents' own financial institutions is significantly higher than general trust in financial institutions, both forms of trust are positively related. Financially knowledgeable people are more likely to trust the prudential supervisor. Finally, our results indicate that trust in the supervisor is positively related to trust in the financial sector.

Keywords: trust, financial institutions, financial literacy, consumer survey. **JEL classifications**: D12, D83, E58, G21, G22.

^{*} Corresponding author. Email: c.a.b.van.der.cruijsen@dnb.nl. We are grateful to CentERdata for collecting the data. The views expressed in this paper do not necessarily reflect the views of DNB or those of the Eurosystem.

1. Introduction

Ten years after the financial crisis, Americans' trust in financial institutions is on the rise, increasing from 22 percent in 2008 to 28 percent in 2018 (Sapienza and Zingales, 2018).¹ Trust in the financial sector may be defined as consumers' expectation that financial institutions are generally dependable and can be relied on to deliver on their promises (Sirdeshmukh et al., 2002). With a high level of trust, customers feel confident that their interests are well served by the financial institution (van Esterik-Plasmeijer and van Raaij, 2017). This is important as financial products are generally imperfectly understood by consumers, particularly because financial products often contain promises about delivering returns far into the future (Jaffer et al., 2014).

Low trust in the financial sector has several potential consequences. First, it may undermine financial stability (Guiso, 2010). In the worst case, it may even lead to bank runs.² Low trust may also damage the financial services industry. If the industry is not trusted, then clients will choose to engage less, which, in turn will damage both the industry and the economy, by reducing the availability of capital for productive purposes (Jaffer et al., 2014). Using survey data from ten Central, Eastern and Southeastern European countries, Stix (2013) reports, for instance, that distrusting people are less likely to have a savings account than trusting people and have stronger liquidity preferences. Finally, low trust may also hurt individual financial institutions as it will make customers less loyal. A loyal customer base contributes to the continuity of financial institutions and less money needs to be spend on attracting new customers (van Esterik-Plasmeijer and van Raaij, 2017). Ampudia and Palligkinis (2018) report, for instance, that Italian households which do not trust the banking sector are less likely to hold a bank account and households are more likely to switch to a new bank if they do not trust their own bank.

This paper analyzes trust in different types of financial institutions in the Netherlands, based on a longitudinal survey. Whereas most previous research (discussed in more detail in section 2), examines one type of financial institution (mostly banks), we analyze trust in multiple types of financial institutions: banks, insurance companies and pension funds. Here we are able to distinguish between trust in respondents' own bank (or insurance company or pension fund) and trust in banks (or insurance companies or pension funds) in general. This is important as previous literature suggests that many customers seem to believe that their own bank is an

¹ Several studies report a decline of trust in banks after the outbreak of the financial crisis (Guiso, 2010; Sapienza and Zingales, 2012; Knell and Stix, 2015). Stevenson and Wolfers (2011) find a significant negative link between the unemployment rate and trust in banks. Personal crisis experiences resulted in a drop of trust in banks in the Netherlands (van der Cruijsen et al., 2016).

² It has been observed that even though people say they don't trust the banks, their behavior suggests otherwise as they neither run nor switch banks frequently. However, the absence of runs probably reflects that protections are in place (like deposit insurance schemes) that make that bank clients feel that their money is safe. Reasons provided for limited switching are that people find it too much of a hassle to change banks, find it a difficult decision to make and are afraid of making mistakes (van der Cruijsen and Diepstraten, 2017), resulting in behavioral loyalty (Dick and Basu, 1994).

exception to the rule that banks cannot be trusted (van Esterik-Plasmeijer and van Raaij, 2017).³ We also examine trust in the integrity and competence of the managers of financial institutions. Several previous studies suggest that these two factors are related to trust in institutions (Pirson and Malhotra, 2008; van Esterik-Plasmeijer and van Raaij, 2017).⁴ Finally, we analyze trust in the financial sector supervisor. Trust in the supervisor has received limited attention in the literature. An exception is the study by van der Cruijsen et al. (2016), who report that personal financial crisis experiences do not have a significant direct effect on trust in the banking supervisor. We hypothesize that trust in the financial sector supervisor enhances trust in the financial sector.

We focus on the role of financial literacy, asking whether more financially educated respondents have more trust in financial institutions in general, their own financial institutions and the supervisor. There is an extensive literature on financial literacy (see Lusardi and Mitchell (2014) for a survey). This literature shows that financially literate people make better financial decisions. For example, they get higher interest rates on their savings accounts (Deuflhard et al., 2019), are more likely to have stocks (van Rooij et al., 2011) and have better diversified portfolios (Von Gaudecker, 2015).⁵ However, little is known on the importance of financial knowledge for trust in financial institutions and the financial sector supervisor. The paper that comes closest to our work is Hansen (2012), who examines the relationship between financial literacy and trust in Danish pension funds and mortgage companies. He finds that consumer knowledge positively influences broad-scope trust (i.e. trust in financial institutions in general) and narrow-scope trust (i.e. trust in someone's own financial institution).⁶ We hypothesize that respondents with good financial knowledge have more trust in their own financial institutions, financial institutions in general and their supervisor. We test this for banks, pension funds and insurance companies.

Our main findings are that, compared to financially illiterate people, financially literate people are more likely to trust financial institutions, their managers, and their supervisor. This holds no matter how we measure financial literacy: based on people's self-assessed financial knowledge, being in charge of household finances or working in the financial sector. We also find that people with a higher degree of trust in the financial health of financial institutions in general (broad-scope trust) are also more likely to trust the financial institutions they are customer of (narrow-scope trust). For all types of financial institutions researched, we find that although the

³ As pointed out by van Esterik-Plasmeijer and van Raaij (2017), customers probably purposefully selected their bank based on their preferences and comparisons with other banks. After this selection, customers are arguably biased and rate their bank as more trustworthy than other banks.

⁴ van Esterik-Plasmeijer and van Raaij (2017) argue that being competent is not enough to be trusted. They define integrity as honesty of employees, fairness in rules, procedures and conditions, and an equal and fair treatment of customers.

⁵ The literature is inconclusive with respect to the effect of financial literacy on bank switching behavior. Brown et al. (2017) and Diepstraten and van der Cruijsen (2019) find that financially literate people are more likely to switch than other consumers, whereas Brunetti et al. (2016) find the opposite.

⁶ In line with these findings, van der Cruijsen and Jonker (2019) report a positive relationship between self-assessed financial knowledge and pensioners' trust in their own pension funds.

latter type of trust is always higher, broad-scope and narrow scope trust are positively related. For all types of institutions, we also find a positive relationship between trust in other people and narrow-scope trust. Furthermore, people with high narrow-scope or broad-scope trust are more likely to trust the integrity and competence of managers of financial institutions than people with low trust in financial institutions. Our results also indicate that trust in the supervisor is positively related to trust in the financial sector. Finally, our findings show that various relationships with sociodemographic variables depend on the type of trust. For example, compared to young people, old people are relatively likely to trust their pension funds but unlikely to trust their banks and life insurance company.

The paper is structured as follows. Section 2 reviews previous studies and formulates our hypotheses. Section 3 describes our data on trust. Section 4 explains the regression method and section 5 presents the results. Section 6 provides conclusions and policy implications.

2. Previous studies and hypotheses

Most studies on trust in financial institutions are single-country studies and focus on banks.⁷ For example, Carbó-Valverde et al. (2013) analyze the drivers of trust in Spanish banks in 2009 and find that it is mainly affected by bank customers' perceived performance of banks, such as their sensitivity to bank customers' problems. Likewise, Jansen et al. (2015) examine which factors may trigger a decline in trust in banks among Dutch consumers by presenting survey respondents with eight hypothetical scenarios related to the financial crisis and asking to what extent these events would harm trust in their banks. A key factor is high executive compensations but other factors such as negative media reports, falling stock prices, and opaque product information can trigger trust loss too. In addition, van der Cruijsen (2019) reports that the commercial usage of payments data can trigger a decline of consumers' trust in their banks.

A few studies employ cross-country data. For instance, Fungáčová et al. (2019) use data on trust in banks from the World Values Survey (WVS) 2010-2014 for 52 countries. They find large differences in trust and show that the level of trust depends on sociodemographic indicators. Females have more trust than males and trust is positively related to income, access to television, being religious, and the holding of pro-market economic views, but is negatively related to education, age and internet access.

Several types of trust have been examined in the literature. Following previous studies (Hansen, 2012; van Esterik-Plasmeijer and van Raaij, 2017), we distinguish between three types

⁷ A few studies focus on trust in Dutch pension funds. van der Cruijsen and Jonker (2019) find that trust in the pension fund's ability to pay pension benefits at all times depends on consumers' perceived pension funds' performance and personal characteristics. van Dalen and Henkens (2018) analyze trust in three types of pension providers: pension funds, banks and insurance companies. Pension funds are trusted most. Trust is positively related to someone's level of education and depends inter alia on the perceived integrity and competence of the pension providers.

of trust: generalized trust, narrow-scope trust and broad-scope trust. Generalized trust (sometimes also called person trust) refers to trust in other people with whom there is no direct relationship. Most studies on generalized trust focus on cross-country comparisons and measure generalized trust as the share of a population answering yes to the following question from the WVS: 'In general, do you think that most people can be trusted, or can't you be too careful in dealing with people?' (see, for instance, Aghion et al. 2010). In our survey (described in more detail in the next section) we use a similar question.

Generalized trust matters for financial decision making at the individual level. For example, people who trust others are more likely to be enrolled in pension plans (Agnew et al., 2007), to become an entrepreneur (Guiso et al., 2006), to participate in the stock market (Balloch et al., 2015), and less likely to default on household debt (Jiang and Lim, 2018). However, there is also research indicating that trust in others can be too high. For instance, Butler et al. (2016) find that when trust very high, people get cheated often and incur large losses.

Following Sirdeshmukh et al. (2002), narrow-scope trust can be defined as consumers' expectation that a specific financial institution is dependable and can be relied on to deliver on its promises. Sometimes this concept is referred to as institution trust (van Esterik-Plasmeijer and van Raaij, 2017). In our research, this concept refers to trust in the health of the bank (or insurance company or pension fund) of the respondent. Broad-scope trust (sometimes also referred to as system trust) is defined as the expectation held by consumers that a group of financial institutions (banks or insurance companies or pension funds) is generally dependable and can be relied on to deliver on their promises (Hansen, 2012). In our research it is measured as trust in the financial health of banks (or pension funds or insurance companies) in general.

We start by formulating our hypotheses about the relationship between the three types of trust. First, we follow van Esterik-Plasmeijer and van Raaij (2017) who argue that the Netherlands has a high level of generalized trust. This is confirmed by data on generalized trust from wave six of the WVS (Inglehart et al., 2014). For example, 66% of the Dutch respondents believe that most others can be trusted, whereas this figure is 35% for US respondents. People usually get information about financial institutions from the mass media, which often report on incidents in the financial sector in a negative way. In contrast, information about a specific bank, insurance company or pension fund is usually obtained from personal experience with this financial institution are often positive (van Esterik-Plasmeijer and van Raaij, 2017). Due to the financial crisis and several scandals, trust in financial institutions (broad-scope trust) decreased, and is expected to be lower than narrow-scope trust. Thus, the first two hypotheses are:

H1. Generalized trust is higher than narrow-scope trust.H2. Narrow-scope trust is higher than broad-scope trust.

van Esterik-Plasmeijer and van Raaij (2017) report support for these hypotheses using a representative sample of 1,079 respondents of 18 years and older from an online consumer panel in the Netherlands. In contrast to the study by van Esterik-Plasmeijer and van Raaij (2017), our study is based on longitudinal data, which therefore captures economic good times and bad times. Furthermore, we test these hypotheses not only for banks, but also for insurance companies and pension funds.

As generalized trust may be related to narrow-scope trust as discussed above, we hypothesize that:

H3. Generalized trust has a positive relationship with narrow-scope trust.

Broad-scope and narrow-scope trust are not independent of each other. Positive personal experiences, satisfaction, and trust with regard to a specific financial institution may be generalized to broad-scope trust, assuming that these institutions are not that different from each other after all. This reasoning suggests that there is a positive relationship between broad-scope and narrow-scope trust. However, van Esterik-Plasmeijer and van Raaij (2017) argue that there may be a negative relationship between broad-scope trust and narrow-scope trust. If broad-scope trust is low, financial institutions may compensate for this by developing narrow-scope trust. Based on this reasoning we formulate the following hypotheses:

H4a. Broad-scope trust has a positive relationship with narrow-scope trust.

H4b. Broad-scope trust has a negative relationship with narrow-scope trust.

van Esterik-Plasmeijer and van Raaij (2017) find that broad-scope trust is strongly and positively related to narrow-scope trust. Hansen (2012) also reports support for Hypothesis 3a, based on surveys among clients of Danish mortgage and pension companies.

In our survey we also asked respondents for their view on the competence and integrity of managers of financial institutions. Competence pertains to several areas of knowledge of managers of financial institutions and their experience as perceived by customers. Integrity can be described as the fairness, morality, and honesty of the managers of financial institutions as perceived by customers. Following van Esterik-Plasmeijer and van Raaij (2017), who find support for these hypotheses, we hypothesize that:

H5a. Trust in the competence and integrity of financial sector managers has a positive relationship with narrow-scope trust.

H5b. Trust in the competence and integrity of financial sector managers has a positive relationship with broad-scope trust.

The relationship between trust in the supervisor and trust in financial institutions has, to the best of our knowledge, never been investigated. This is rather surprising in view of the fact that strengthening trust in the financial sector is an important objective in the mission statements of many supervisory agencies. Supervisory authorities are aiming to make financial institutions safer and the financial system more stable and resilient. If the public trust that the supervisory authorities can deliver on these objectives, their trust in the financial sector may also be higher. We therefore hypothesize:

H6. Trust in the financial sector supervisor is positively related to broad-scope trust. Finally, we turn to the role of financial literacy. Past research suggests that knowledgeable consumers acquire and retain more information than less knowledgeable consumers (see Hansen (2012) for a discussion of this literature). Knowledgeable consumers may be more able to detect the limitations of a financial service provider, thus decreasing trust. However, financial knowledge is likely to make consumers more aware of what financial service providers do and this may lead to more trust. Indeed, van der Cruijsen and Jonker (2019) report that trust in pensions is higher among pensioners who self-assess to be more or less knowledgeable with respect to financial matters than for pensioners who consider themselves not knowledgeable. Hansen (2012) also reports that consumer knowledge positively influences narrow- and broad-scope trust. We therefore expect that:

H7. Financial knowledge has a positive relationship with broad-scope trust.

H8. Financial knowledge has a positive relationship with narrow-scope trust.

For similar reasons, better financial knowledge may affect trust in financial sector managers and the financial sector supervisor. We therefore hypothesize:

H9. Financial knowledge has a positive relationship with trust in the competence and integrity of financial sector managers.

H10. Financial knowledge has a positive relationship with trust in the financial sector supervisor.

Finally, we test a hypothesis put forward by Hansen (2012) who argues that when broad-scope trust is low, consumers will rely more on their financial knowledge in determining the trust-worthiness of their own financial service provider than when broad-scope trust is high. In other words:

H11. Financial knowledge has a greater positive effect on narrow-scope trust when broadscope trust is low than when broad-scope trust is high.

Hansen (2012) finds some weak evidence that broad-scope trust negatively moderates the relationship between consumers' financial knowledge and narrow-scope trust. This implies that financial knowledge has a greater positive effect on narrow-scope trust when broad-scope trust is low rather than high.

2. Data on trust

2.1 DNB Trust Survey and DNB Household Survey

Each year, De Nederlandsche Bank (DNB) collects detailed data on consumers' trust in the financial sector via the DNB Trust Survey (DTS). The DTS is held among the CentERpanel, a representative sample of the Dutch-speaking population in the Netherlands.⁸ This internet-based panel consists of approximately 2,000 households. All family members aged 16 and above in the panel are invited to complete the DTS. The DTS includes a question to measure trust in DNB (DNB is responsible for supervising banks, pension funds and insurance companies), and also a question on trust in other people (generalized trust). Moreover, it includes questions that zoom in on trust in the financial health of various types of financial institutions, and the perceived competence and integrity of the managers of these institutions. Many of the questions have been part of the DTS since its start and have remained unchanged. We use data from 2006 until 2019.⁹ This enables us to track the evolvement of trust. An important advantage of our data is that it can easily be linked to data on personal characteristics and financial knowledge. This information is captured by the annual DNB Household Survey (DHS), which is also filled-in by members of the CentERpanel. The DHS exists already since 1993 and has been extensively used by researchers and policymakers because it covers a wide range of topics.¹⁰ Prior research has shown that usage of the DTS can result in valuable new insights (Mosch and Prast, 2008; Jansen et al., 2015; van der Cruijsen et al., 2016; Diepstraten and van der Cruijsen, 2019; van der Cruijsen et al., 2019; van der Cruijsen, 2019).

2.2 Generalized trust, trust in the banking supervisor, and trust in the financial sector

Generalized trust is elicited by asking 'Generally speaking, would you say that most people can be trusted, or that you can't be too careful in dealing with people?'. This question is very similar to the generalized trust question in the WVS and the US General Social Survey. Respondents choose between 'people can be trusted' and 'one cannot be careful enough'. In 2019, 63% of the respondents think that, in general, most people can be trusted (Figure 1). The share of respondents with trust in other people was the lowest in 2006 (60%) and the highest in 2011 (70%). Over the entire sample, the share of respondents with trust in other people was 65%. In regressions we include the dummy *generalized trust*, which is one if people answer that most people can be trusted and which is zero otherwise.

⁸ The CentERpanel is managed by CentERdata, a research institute affiliated with Tilburg University. For more information on this panel we refer to Teppa and Vis (2012).

⁹ Two more general measures of trust in the financial sector are part of our survey data but only since 2019. These measures are used in our robustness analysis.

¹⁰ A list with publications using data collected among the CentERpanel is available on https://www.centerdata.nl/en/publications. See http://www.centerdata.nl/en/projects-by-centerdata/dnb-household-survey-dhs for more information on the DHS. URLs have been last accessed on November 28, 2019.

Trust in the supervisor is an ordered variable capturing trust in DNB. It ranges from 1 (absolutely no trust) to 4 (a lot of trust). Figure 1 shows that trust in the supervisor declined sharply during the financial crisis and has not completely recovered yet. In 2019, trust in the supervisor was 2.8 on average.



Figure 1. Trust in the supervisor (DNB) and generalized trust

Source: DTS.

Note: The figure reports average levels of trust with 95% confidence intervals.

The DTS includes questions to measure trust in the financial soundness of three types of financial institutions: banks, insurance companies and pension funds. People are asked whether they trust, at all times, banks to repay their deposits, insurance companies to pay insurance money and pension funds to pay pension benefits. The questions asked refer to financial institutions in general (broad-scope trust measures) and to the respondents' own financial institutions (narrow-scope trust measures).

Narrow-scope trust in banks measures the trust in one's own bank(s) capacity to repay one's deposits at all times. *Broad-scope trust in banks* measures trust in the ability to repay of all banks in the Netherlands. *Narrow-scope trust in life insurance company* captures trust in one's own life insurance company that it pays for insurance at all times. *Broad-scope trust in insurance companies* measures trust in the fulfilment of payment obligations to all persons by all type of insurance companies, also at all times. In a similar way *narrow-scope trust in pension funds* and *broad-scope trust in pension funds* measure trust in pension funds' ability to pay pension benefits at all times, either one's own pension benefit (the first variable) or all pension benefits (the second variable). *Narrow-scope trust: average* is the average of the narrow-scope trust measures for the different types of institutions. For each respondent this average is based on the available narrow-scope trust measures, so at most three. For example, if we only have narrow-scope trust in banks and narrow-scope trust in pension funds for a respondent, narrow-scope trust: average is simply the average of these two measures. *Broad-scope trust: average* is the average of the broad-scope trust measures and constructed in a similar way as narrow-scope trust: average.

Nowadays, trust in the health of financial institutions is lower than fourteen years ago (Figure 2). It declined during the financial crisis and has not fully recovered yet. Trust in different types of financial institutions is on average between 3 (neutral) and 4 (yes, predominantly).

The 2006-2019 average trust in the financial health of one's own bank was 4.0. This is significantly higher than the average narrow-scope trust in pension funds, which was 3.7 (t-statistic: 46.9). Narrow-scope trust in life insurance companies was 3.9, so also higher than narrow-scope trust in pension funds (t-statistic: 26.8). We observe higher trust in life insurance companies for each year, except for 2019 (when there is no significant difference).

Average broad-scope trust in banks was also higher than average broad-scope trust in pension funds: 3.7 versus 3.4 (t-statistic: 56.5). Insurance companies could also count on more broad-scope trust than pension funds (t-statistic: 47.5). Only in 2008 there was no significant difference. On average, broad-scope trust in insurance companies was 3.6. Although on average broad-scope trust in banks is somewhat higher than trust in insurance companies, the difference is only 0.1 (t-statistic: 10.6).

The data underlying Figure 2 allow us to test H1. Table 1 shows the results of paired ttests. The figures for average generalized trust differ in the columns due to differences in the number of respondents answering the questions on generalized trust and narrow-scope trust for different types of financial institutions. For banks and life insurance companies we reject H1: generalized trust is lower than narrow-scope trust.¹¹ In contrast, for pension funds we find support for H1.

We also test for differences between generalized trust and narrow-scope trust for each year in the sample. The findings support our prior findings for banks but these tests yield mixed outcomes in case of life insurances and pension funds (see Appendix B, Table B.1).

¹¹ Note that we had to construct a somewhat different measure for generalized trust to simplify the comparison with the narrow-scope trust measures (which run from 1 to 5). In Table 1, *generalized trust 2* is 1 if generalized trust is 0 and it is 5 if generalized trust is 1.



Figure 2. Trust in the financial health of financial institutions and generalized trust

Source: DTS.

Note: The figures report average levels of trust with 95% confidence intervals. From 2006-2016 all respondents answered the question about broad-scope trust in insurance companies, whereas in 2018 and 2019 the question was only answered by respondents with a life insurance.

	Banks	Life insurance company	Pension funds
(a) Generalized trust 2	3.64	3.67	3.74
(b) Narrow-scope trust	3.95	3.92	3.68
(a) - (b)	-0.32***	-0.25***	0.06***
T-statistic	-28.86	-15.05	4.92
Number of observations	30,660	13,219	22,382

Table 1. The difference between generalized trust and narrow-scope trust

Note: *** denotes statistical significance at the 0.01 level (one-sided paired t-test). Period: 2006-2019.

In general, trust in the financial health of the own institution is higher than trust in the health of financial institutions in general (Figure 2 and Table 2). This provides support for H2. The t-statistics for the difference between broad-scope and narrow scope trust over the entire sample period are 70.5 for banks, 32.6 for insurance companies and 53.5 for pension funds. We also find support for H2 when we test the difference for each year separately (Appendix B, Table B.2).

	=	_	
	Banks	(Life) insurance companiesª	Pension funds
(a) Narrow-scope trust	3.96	3.93	3.68
(b) Broad-scope trust	3.70	3.76	3.46
(a) - (b)	0.26***	0.17***	0.22***
T-statistic	70.54	32.64	53.48
Number of observations	30,358	13,093	22,213

Table 2. The difference between narrow-scope trust and broad-scope trust

Note: *** denotes statistical significance at the 0.01 level (one-sided paired t-test). Period: 2006-2019. a Narrow-scope trust is trust in one's own life insurance company, whereas broad-scope trust captures general trust in all insurance companies.

Figure 3 shows trust in the competence and integrity of managers of financial institutions, which is measured separately as of 2010. *Trust in managers' competence and integrity* measures to what extent respondents agree with the statement: 'Managers of financial institutions are in general knowledgeable and sound'. This ordered variable ranges from 1 (completely disagree) to 5 (completely agree). *Trust in managers' competence* and *trust in managers' integrity* are built in a similar fashion but only focus on one of the two characteristics of managers. On average trust in the competence is 3.2 and trust in the integrity is 2.8, so 0.4 lower (t-statistic: 80.4). In 2019, 23% of the respondents disagreed or strongly disagreed with the statement that managers of financial institutions are integer, 25% (strongly) agreed, 44% took a neutral stance and 9% did not know what to answer. Regarding competence, 10% think managers are incompetent, 46% perceive them as competent, 34% have a neutral standpoint and 9% did not report an opinion.



Figure 3. Trust in the competence and integrity of managers of financial institutions

Source: DTS.

Note: The figure reports average levels of trust with 95% confidence intervals.

3. Regression method

We run various regressions to test our other hypotheses. In this section we explain the construction of the variables included in these models (apart from the trust variables as discussed in section 2) and the models themselves.

3.1 Financial knowledge

Financial knowledge captures self-assessed knowledge of financial matters and can take four different values: 1 = not knowledgeable, 2 = more or less knowledgeable, 3 = knowledgeable, or 4 = very knowledgeable. Unfortunately, 2019 data on self-assessed knowledge of financial matters is not available. Therefore, we take the most recent available data, assuming that the self-assessment did not change. Van Rooij et al. (2011) show a very strong link between a self-reported financial knowledge measure and literacy measures based on knowledge questions.

3.2 Control variables

We include a wide range of control variables. *Male* is a binary dummy that is 1 for males and 0 for females. Four binary *age dummies* capture the age of the respondent: between 35 and 44, between 45 and 54, between 55 and 64, 65 and over. For example, between 35 and 44 is 1 for respondents between 35 and 44 and 0 for respondents younger than 35 or older than 44. Respondents of 34 years and below are in the reference category. *Education: bachelor or higher* is 1 for respondents who successfully completed higher vocational or university education and 0 for lower-educated respondents. Three binary *income dummies* are constructed to control for differences in the

household net monthly income: EUR 1151-1800, income: EUR 1801-2600, income: > EUR 2600. The dummies are 1 for respondents with the particular income and 0 for respondents with another level of income. Respondents in the reference category have an income of EUR 1150 or below. We also control for having a job: the binary dummy job is 1 for respondents who have a paid job, work in family business or are self-employed and 0 for other respondents. Homeowner is included as a proxy for wealth. This variable is 1 for homeowners and 0 else. If the household head lives together with a partner the variable household head lives with partner is 1 and otherwise it is 0. *Degree of urbanization* ranges from 1 (the address density of the respondent's residence is 500 per km² or less) to 5 (address density of more than 2500 per km²). We also control for the region people live in by including the binary region dummies north, east, and south. These variables are 1 for respondents who live in the mentioned region and 0 else. Respondents who live in the west of the Netherlands are in the reference group. Last, we include *risk aversion*. It is the average agreement measured on a 1 (totally disagree) to 7 (totally agree) scale with six statements on risk taking. For example, the statement 'I think it is more important to have safe investments and guaranteed returns, than to take a risk to have a chance to get the highest possible returns'.

In addition, we control for personal crisis experiences. In line with van der Cruijsen et al. (2016) we use the 2010 and 2013 DTS information on personal crisis experiences. The March 2010 DTS included a couple of questions on personal crisis experiences. These questions measure whether respondents had savings at one of the banks that either received government support or went bankrupt in 2008/2009. The March 2013 DTS included questions on respondents' crisis experiences with respect to the nationalization of SNS Reaal in 2013. It was asked whether respondents had savings at one of the banks that were part of SNS Reaal (ASN Bank, SNS Bank, and/or RegioBank). We use this information to construct two binary dummy variables: year after bankruptcy and year after bailout. Year after bankruptcy is 1 in the year after a bankruptcy for customers who experienced that their bank went bankrupt in the prior year. So, it is 1 in 2009 for respondents who had savings at Icesave in 2008 and it is 1 in 2010 for respondents who had savings at DSB in 2009. Year after bailout is 1 in 2009 for respondents who were customer of a bank that was bailed out in 2008 and 1 in 2013 for respondents who were customer of ASN Bank, SNS Bank, and/or RegioBank. In all other cases it is 0. Table A.1 in Appendix A provides more detailed information on the variable construction and the descriptive statistics of all variables used.

3.3 Models

We estimate several panel models. As all dependent variables are ordered variables, we estimate random-effects ordered logistic regressions. First, we estimate a model with broad-scope trust_{i,f,t}

as dependent variable (model 1a). This model enables us to test whether there is a positive link between financial knowledge_{i,t} and broad-scope trust_{i,f,t} (H7). We run the model for each type of financial institution separately.

broad-scope trust_{i,f,t} =
$$f(financial knowledge_{i,t}, X_{i,t}) + u_i + e_{i,t}$$
 (1a)

In this and all subsequent equations f denotes the type of financial institution (either banks, insurance companies or pension funds), i indicates the individual, and t refers to time. Financial knowledge_{i,t} is the self-assessed financial knowledge and the key explanatory variable (but, as explained in more detail below, we also experiment with other proxies for financial knowledge). The vector $X_{i,t}$ captures personal characteristics. It also includes personal crisis experiences in case f=banks. The error term is composed of an idiosyncratic error $e_{i,t}$ and a household fixed component u_i which controls for unobserved heterogeneity.

Equation 1b is the same as equation 1a but with trust in the supervisor_{i,t} as additional explanatory variable. This enables us to test whether there is a positive relationship between trust in the supervisor and broad-scope trust (H6).

broad-scope trust_{*i*,*f*,*t*} = $f(financial knowledge_{i,t}, X_{i,t} trust in the supervisor_{i,t}) + u_i + e_{i,t}$ (1b)

Next, we run a model with trust in the supervisor_{i,t} as dependent variable to test whether there is a positive relationship between financial knowledge and trust in DNB (H10). We run two regressions, one without and one with the personal crisis experiences included in X_{i,t}.

trust in the supervisor_{i,t} =
$$f(financial knowledge_{i,t}, X_{i,t}) + u_i + e_{i,t}$$
 (2)

Thereafter, we estimate a set of models with narrow-scope trust_{i,f,t} as dependent variable. We run models 3a, 3b and 3c for each type of financial institution separately.

narrow-scope trust _{i,f,t} = $f(financial knowledge_{i,t}, X_{i,t}) + u_i + e_{i,t}$	(3a)
narrow-scope trust _{i,f,t} = $f(financial knowledge_{i,t}, X_{i,t}, broad-scope trust_{i,f,t}) + u_i + e_{i,t}$	(3b)
narrow-scope trust _{i,f,t} = f(financial knowledge _{i,b} $X_{i,b}$ broad-scope trust _{i,f,t} ,	
financial knowledge _{i,t} *broad-scope trust _{i,f,t}) + u _i + e _{i,t}	(3c)

Equation 3a enables us to test whether financial knowledge has a positive relationship with narrow-scope trust (H8). In case f=banks the vector $X_{i,t}$ includes personal crises experiences. In one version of this equation, the vector $X_{i,t}$ includes generalized trust (as a dummy variable),

which allows us to test H3. Equation 3b is an extended version of equation 3a (with generalized trust); broad-scope trust_{i,f,t} is added to test H4a and H4b. Equation 3c also includes the interaction term financial knowledge_{i,t}*broad-scope trust_{i,f,t} to test whether financial knowledge has a greater positive effect on narrow-scope trust when broad-scope trust is low than when broad-scope trust is high (H11).

Finally, we estimate a couple of models with trust in managers' competence and integrity_{i,t} as dependent variable using data on 2006-2009. For the period 2010-2019 we are able to estimate two sets of regressions, one with trust in managers' competence_{i,t} as dependent variable and one with trust in managers' integrity_{i,t} as dependent variable. Equation 4a is estimated to test H9: the presence of a positive link between financial knowledge and trust in the competence and integrity of managers of financial institutions. Model 4b includes narrow-scope trust: average_{i,t} to test whether there is a positive relationship between narrow-scope trust and trust in the competence and integrity of managers of financial institutions (H5a). Model 4c includes broad-scope trust: average_{i,t} instead to test H5b. In all these equations the vector $X_{i,t}$ includes personal crises experiences.

trust in managers' competence and integrity_{i,t} =
$$f(financial knowledge_{i,t}, X_{i,t}) + u_i + e_{i,t}$$
 (4a)
trust in managers' competence and integrity_{i,t} = $f(financial knowledge_{i,t}, X_{i,t})$

*narrow-scope trust: average*_{*i*,*t*}) +
$$u_i$$
 + $e_{i,t}$ (4b)

trust in managers' competence and integrity_{i,t} = $f(financial knowledge_{i,t}, X_{i,t}, broad-scope trust: average_{i,t}) + u_i + e_{i,t}$ (4c)

4. Results

4.1 Financial knowledge is key for broad-scope trust

We find convincing support for H7: financial knowledge has a positive relationship with broadscope trust. Table 3 reports the regression results of model 1a (see column 1a, 2a and 3a) and model 1b that includes trust in the supervisor (see column 1b, 2b, and 3b). The coefficient on financial knowledge is positive and significant for all three types of broad-scope trust. People who consider themselves to be very knowledgeable of financial matters are 3 percentage points more likely to predominantly trust banks and also 3 percentage points more likely to completely trust banks than people who think they are unknowledgeable. In case of insurance companies these effects are 3 and 2 percentage points, whereas they are 3 and 1 percentage points in case of pension funds.

	Broad-scope trust in		Broad-sco	pe trust in	Broad-scope trust in			
	ba	nks	insurance	companies	pensio	n funds		
	(1a)	(1b)	(2a)	(2b)	(3a)	(3b)		
Financial knowledge	0.13***	0.13***	0.11***	0.11***	0.10***	0.09***		
	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)		
Male	0.36***	0.32***	0.36***	0.33***	0.38***	0.35***		
	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)		
Between 35 and 44	-0.08	-0.06	-0.02	0.00	0.19**	0.23***		
	(0.09)	(0.09)	(0.08)	(0.08)	(0.08)	(0.08)		
Between 45 and 54	0.01	0.01	-0.18**	-0.19**	0.36***	0.38***		
	(0.09)	(0.09)	(0.09)	(0.09)	(0.08)	(0.08)		
Between 55 and 64	-0.05	-0.02	-0.35***	-0.33***	0.52***	0.56***		
	(0.09)	(0.09)	(0.09)	(0.09)	(0.09)	(0.09)		
65 and over	-0.14	-0.10	-0.44***	-0.41***	0.55***	0.61***		
	(0.10)	(0.10)	(0.10)	(0.10)	(0.10)	(0.09)		
Education: bachelor or hiaher	0.02	0.02	0.07	0.07	0.06	0.06		
	(0.07)	(0.07)	(0.07)	(0.07)	(0.07)	(0.07)		
Income: EUR 1151-1800	0.23*	0.21*	0.08	0.06	0.47***	0.44***		
	(0.12)	(0.12)	(0.12)	(0.12)	(0.12)	(0.12)		
Income: EUR 1801-2600	0.22*	0.21*	0.01	0.01	0.32**	0.31**		
	(0.12)	(0.12)	(0.12)	(0.12)	(0.13)	(0.12)		
Income: > EIIR 2600	0.09	0.11	0.00	0.02	0.27**	0.28**		
	(0.13)	(0.13)	(0.13)	(0.13)	(0.13)	(0.13)		
loh	0.06	0.06	0.10	0.10	0.05	0.04		
<u>j</u> 00	(0.07)	(0.07)	(0.07)	(0.07)	(0.03)	(0.07)		
Homeowner	0.01	0.01	0.13*	0.13*	0.08	0.08		
nomeowner	(0.01)	(0.08)	(0.08)	(0.08)	(0.08)	(0.08)		
Household head lives with partner	0.08	0.07	0.10	0.07	0.16**	0.14*		
nousenolu neuu nves with purther	(0.08)	(0.08)	(0.08)	(0.08)	(0.08)	(0.08)		
Dearge of urbanization	0.01	0.01	0.00	0.00	0.03	0.03		
	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)		
Region: north	0.09	0.12	-0.07	-0.05	0.04	0.07		
Region. nor th	(0.12)	(0.12)	(0.12)	-0.03	(0.11)	(0.11)		
Pagion: agst	0.12	0.12	0.12	(0.12)	0.12	.0.11		
Keylon. eust	-0.13	-0.13	-0.11	(0.09)	-0.13	-0.11		
Pagion, south	0.10	0.09	0.09)	0.05	(0.09)	0.05		
Keylon. south	-0.10	-0.00	-0.08	-0.00	(0.0)	-0.03		
Dick quarsian	0.09	(0.09)	0.05	0.05	(0.09)	(0.09)		
RISK UVEI SION	-0.01	(0.00)	(0.02)	0.01	-0.03	-0.02		
Voar after hankrunten	0.12	0.10	(0.02)	(0.02)	(0.02)	(0.02)		
Tear after bankrupicy	-0.13	-0.19						
Voor after hailent	(0.19)	(0.10)						
reur ajter ballout	-0.50	-0.39						
Transfirm the same articles	[0.09]	(U.U9) 0.40***		0 10***		0 53***		
Trust in the supervisor		0.48		0.40		0.53		
		(0.03)		(0.03)		(0.03)		
Number of chapterstices	22047	22.044	21.040	21.046	22 7/1	22 750		
Number of respondents	23,047 E 204	23,044 E 204	41,747 E 222	21,740 E 222	22,/01 E 247	22,/30 E 247		
Wold v^2	3,374 101 01***	0,074 102 00***	0,002 110 60***	3,332 240 00***	3,347 156 96***	3,347 604 05***		
vvalu X-	TOT'AT	404.70	112.02	347.07	120.00	004.73		

Table 3. Financial knowledge and broad-scope trust

Note: The table reports parameter estimates of random effects ordered logit regressions. Period: 2006-2019. Standard errors are clustered by household and shown in parentheses. The dependent variables range from 1 (no trust at all) to 5 (complete trust). ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively.

The results also confirm our expectation of a positive relationship between trust in the supervisor and broad-scope trust (H6). The effect of trust in the supervisor on broad-scope trust is strong. For example, in case of banks the effect implies that people who have a lot of trust in DNB are 10 percentage more likely to predominantly trust banks and also 10 percentage points more likely to completely trust banks than people who have absolutely no trust in DNB.

In addition, broad-scope trust is related to sociodemographic variables. We find that males trust all types of institutions more than females. For example, in case of banks the likelihood that males answer that they predominantly trust banks is 2 percentage points higher. The same holds for the likelihood of answering 'complete trust'. Negative age effects are present in case of broadscope trust in insurance companies, whereas trust in pension funds positively depends on age. For example, people aged 65 and over are 10 percentage points more likely to predominantly or completely trust their pension funds than people in the reference group. Trust in pension funds is higher for people with an income above EUR 1150 than for people with less income. In case of banks, there are also some positive income effects, although they are weaker. We find no significant relationship between income and broad-scope trust in insurance companies. Homeowners have more trust in insurance companies than renters. People who are part of a household in which the household head has a partner are more likely to trust pension funds than people who live in another type of household. All types of trust are unrelated to the level of education, having a job, the degree of urbanization of people's place of residence, the region where people live and risk aversion. Confirming the findings of van der Cruijsen et al. (2016), our results suggest that personal crisis experience matter. Broad-scope trust is lower for people who experienced in the prior year that their bank was bailed-out. For example, people are 5 percentage points less likely to predominantly or completely trust banks in the year after they experienced a bail-out of their bank than people without such a personal crisis experience.

4.2 Financial knowledge also matters for trust in the supervisor

We also find support for H10: financial knowledge is significantly positively related to trust in the prudential banking supervisor DNB. Table 4 shows the estimation results for model 2. People who say they are very knowledgeable of financial matters are 3 percentage points more likely to have a lot of trust in the supervisor than people who think they are not financially knowledgeable. Very financially knowledgeable people are also 3 percentage points more likely to have pretty much trust in the supervisor. Column 2 shows the results of a regression that includes controls for personal crisis experiences. Surprisingly, trust in DNB is significantly higher for people who experienced a bankruptcy or bailout of their bank.

	Trust in the	supervisor
	(1a)	(1b)
Financial knowledge	0.09***	0.09***
	(0.02)	(0.02)
Male	0.21***	0.21***
	(0.03)	(0.03)
Between 35 and 44	-0.08	-0.08
	(0.06)	(0.06)
Between 45 and 54	0.08	0.08
	(0.06)	(0.06)
Between 55 and 64	0.01	-0.00
	(0.06)	(0.06)
65 and over	0.01	0.01
	(0.07)	(0.07)
Education: bachelor or higher	0.07*	0.06
	(0.04)	(0.04)
Income: EUR 1151-1800	0.17**	0.17**
	(0.08)	(0.08)
Income: FIIR 1801-2600	0.09	0.09
Income. Box 1001 2000	(0.08)	(0.08)
Income: > FIIR 2600	-0.00	0.00
meome. > Eon 2000	(0.09)	(0.09)
Ioh	(0.07)	0.02
100	-0.02	-0.02
Homeourge	(0.04)	0.04)
Homeowner	0.09 ⁻	(0.05)
Household hand lives with a suture	(0.05)	(0.03)
Househola neda lives with partner	0.05	0.04
	(0.05)	(0.05)
Degree of urbanization	0.00	0.00
	(0.02)	(0.01)
Region: north	-0.08	-0.08
	(0.06)	(0.06)
Region: east	-0.13**	-0.12**
	(0.05)	(0.05)
Region: south	-0.13***	-0.13***
	(0.05)	(0.04)
Risk aversion	-0.02	-0.03
	(0.02)	(0.02)
Year after bankruptcy		0.50***
		(0.18)
Year after bailout		0.16**
		(0.08)
Number of observations	23 719	23 541
Number of respondents	5 479	5 4 5 2
Wald v2	125 45***	138 28***
	145.15	100.20

Table 4. Financial knowledge and trust in the supervisor

Note: The table reports parameter estimates of random effects ordered logit regressions. Period: 2006-2019. Standard errors are clustered by household and shown in parentheses. The dependent variables range from 1 (absolutely no trust) to 4 (a lot of trust). ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively.

4.3 Financial knowledge is also important for narrow-scope trust

Our findings on narrow-scope trust support H8: people with a higher degree of financial knowledge are more likely to trust their own financial institutions. Table 5 focusses on trust in the financial soundness of respondents' own banks, insurance companies and pension funds and shows that a higher degree of financial knowledge also goes along with more trust in the financial health of these institutions. Compared to financially unknowledgeable people, very financially knowledgeable people are more likely to trust banks to be able to repay their deposits at all times

	Na	irrow-scope	e trust in bar	ıks	Narrow-s	scope trust i	n insurance	ce company Narrow-scope trust in pension funds					
	(1a)	(1b)	(1c)	(1d)	(2a)	(2b)	(2c)	(2d)	(3a)	(3b)	(3c)	(3d)	
Financial knowledge	0.09***	0.09***	0.06**	-0.09	0.10**	0.10**	0.07	-0.37*	0.10**	0.10**	0.07**	0.01	
C C	(0.03)	(0.03)	(0.03)	(0.15)	(0.05)	(0.05)	(0.04)	(0.22)	(0.04)	(0.04)	(0.03)	(0.13)	
Male	0.46***	0.47***	0.30***	0.30***	0.51***	0.52***	0.27***	0.27***	0.57***	0.59***	0.34***	0.34***	
	(0.06)	(0.06)	(0.05)	(0.05)	(0.08)	(0.08)	(0.07)	(0.07)	(0.07)	(0.07)	(0.06)	(0.06)	
Between 35 and 44	-0.22***	-0.22***	-0.23***	-0.23***	-0.30***	-0.32***	-0.29***	-0.29***	0.21**	0.21**	0.09	0.09	
	(0.08)	(0.08)	(0.07)	(0.07)	(0.11)	(0.11)	(0.10)	(0.10)	(0.10)	(0.10)	(0.08)	(0.08)	
Between 45 and 54	-0.18**	-0.19**	-0.22***	-0.22***	-0.52***	-0.54***	-0.43***	-0.43***	0.44***	0.43***	0.24***	0.24***	
	(0.09)	(0.09)	(0.07)	(0.07)	(0.12)	(0.12)	(0.11)	(0.11)	(0.11)	(0.11)	(0.08)	(0.08)	
Between 55 and 64	-0.26***	-0.24***	-0.20***	-0.20***	-0.56***	-0.59***	-0.32***	-0.32***	0.85***	0.85***	0.78***	0.78***	
	(0.09)	(0.09)	(0.08)	(0.08)	(0.13)	(0.13)	(0.12)	(0.12)	(0.11)	(0.11)	(0.09)	(0.09)	
65 and over	-0.33***	-0.29***	-0.13	-0.13	-0.77***	-0.78***	-0.50***	-0.50***	1.00***	1.02***	0.93***	0.93***	
	(0.10)	(0.10)	(0.08)	(0.08)	(0.15)	(0.15)	(0.14)	(0.14)	(0.13)	(0.13)	(0.10)	(0.10)	
Education: bachelor or higher	0.07	-0.04	0.02	0.02	-0.03	-0.13	-0.00	-0.00	-0.06	-0.13*	-0.08	-0.08	
	(0.07)	(0.07)	(0.05)	(0.05)	(0.09)	(0.09)	(0.07)	(0.07)	(0.08)	(0.08)	(0.06)	(0.06)	
Income: EUR 1151-1800	0.17	0.17	0.11	0.10	0.16	0.17	0.20	0.20	0.23	0.24	-0.04	-0.04	
	(0.13)	(0.12)	(0.11)	(0.11)	(0.21)	(0.21)	(0.21)	(0.21)	(0.15)	(0.15)	(0.13)	(0.13)	
Income: EUR 1801-2600	0.12	0.10	0.01	0.01	0.18	0.15	0.24	0.24	0.19	0.18	0.07	0.07	
	(0.13)	(0.13)	(0.11)	(0.11)	(0.21)	(0.21)	(0.21)	(0.21)	(0.15)	(0.15)	(0.13)	(0.13)	
Income: > EUR 2600	-0.05	-0.09	-0.07	-0.07	0.13	0.08	0.19	0.18	0.18	0.16	0.16	0.16	
	(0.14)	(0.14)	(0.12)	(0.12)	(0.22)	(0.22)	(0.21)	(0.21)	(0.16)	(0.16)	(0.13)	(0.13)	
Job	0.03	0.02	-0.03	-0.03	-0.15*	-0.17*	-0.22***	-0.22***	-0.08	-0.08	-0.17**	-0.17**	
	(0.07)	(0.07)	(0.06)	(0.06)	(0.09)	(0.09)	(0.09)	(0.09)	(0.08)	(0.08)	(0.07)	(0.07)	
Homeowner	0.01	-0.04	-0.00	-0.00	0.24*	0.18	0.09	0.09	-0.00	-0.03	-0.03	-0.03	
	(0.08)	(0.08)	(0.07)	(0.07)	(0.13)	(0.12)	(0.11)	(0.11)	(0.09)	(0.09)	(0.07)	(0.07)	
Household head lives with partner	0.11	0.13	0.06	0.06	0.28**	0.30**	0.25**	0.24**	0.08	0.10	0.01	0.01	
	(0.08)	(0.08)	(0.07)	(0.07)	(0.12)	(0.12)	(0.10)	(0.10)	(0.09)	(0.09)	(0.07)	(0.07)	
Degree of urbanization	0.01	0.01	0.01	0.01	-0.02	-0.02	-0.02	-0.02	0.03	0.03	0.02	0.02	
	(0.03)	(0.03)	(0.02)	(0.02)	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)	(0.02)	(0.02)	
Region: north	0.04	0.01	-0.02	-0.02	-0.01	-0.03	0.03	0.03	0.02	0.01	0.01	0.01	
	(0.12)	(0.12)	(0.09)	(0.09)	(0.15)	(0.15)	(0.12)	(0.12)	(0.13)	(0.13)	(0.10)	(0.10)	
kegion: east	-0.15	-0.15	-0.06	-0.06	-0.14	-0.16	-0.08	-0.08	-0.06	-0.06	-0.02	-0.02	
Design anth	(0.10)	(0.10)	(0.08)	(0.08)	(0.12)	(0.12)	(0.09)	(0.09)	(0.11)	(0.10)	(0.08)	(0.08)	
kegion: south	-0.14	-0.15	-0.11	-0.11	0.00	-0.01	0.01	0.02	-0.08	-0.08	-0.03	-0.03	
	(0.09)	(0.09)	(0.07)	(0.07)	(0.12)	(0.11)	(0.10)	(0.10)	(0.10)	(0.10)	(0.07)	(0.07)	
KISK AVERSION	0.06**	0.06**	0.11^{***}	0.11^{***}	0.05	0.05	0.09***	0.09***	-0.01	-0.01	0.04*	0.04*	
	(0.02)	(0.02)	(0.02)	(0.02)	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)	(0.02)	(0.02)	

Table 5. Financial knowledge and narrow-scope trust

Table 5. Financial knowledge and narrow-scope trust (cor	it.)
--	------

	Narrow-scope trust in banks Nar			Narrow-	scope trust i	n insurance d	company	Narrow-scope trust in pension funds				
	(1a)	(1b)	(1c)	(1d)	(2a)	(2b)	(2c)	(2d)	(3a)	(3b)	(3c)	(3d)
Year after bankruptcy	-0.18	-0.13	-0.09	-0.09								
	(0.19)	(0.20)	(0.21)	(0.21)								
Year after bailout	-0.50***	-0.53***	-0.35***	-0.35***								
	(0.09)	(0.09)	(0.10)	(0.10)								
Generalized trust		0.77***	0.53***	0.53***		0.66***	0.27***	0.27***		0.51***	0.39***	0.39***
		(0.05)	(0.05)	(0.05)		(0.07)	(0.07)	(0.07)		(0.06)	(0.05)	(0.05)
Broad-scope trust in banks			2.64***	2.55***								
			(0.05)	(0.10)								
Financial knowledge*broad-scope trust in				0.04								
banks				(0.04)								
Broad-scope trust in insurance companies							3.08***	2.83***				
-							(0.08)	(0.14)				
Financial knowledge*broad-scope trust in								0.12**				
insurance companies								(0.06)			2 2 2 * * *	2 1 0 * * *
Broad-scope trust in pension funds											3.22***	3.18***
Financial Imoulodoc*broad coope truct in											(0.06)	(0.10)
rinancial knowledge broad-scope crust in												(0.02)
pension junus												(0.04)
Number of observations	23,181	23,181	22,961	22,961	9,818	9,818	9,735	9,735	17,630	17,630	17,513	17,513
Number of respondents	5,410	5,410	5,389	5,389	3,347	3,347	3,326	3,326	4,389	4,389	4,377	4,377
Wald χ^2	130.92***	370.37***	2995.83***	3002.29***	94.01***	171.69***	1706.28***1	708.59***	234.53***	323.27***	3555.14***	3557.26***

Note: The table reports parameter estimates of random effects ordered logit regressions. Period: 2006-2019. Standard errors are clustered by household and shown in parentheses. The dependent variables range from 1 (no trust at all) to 5 (complete trust). ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively.

(column 1a), life insurance companies to pay insurance money at all times (column 2a), and pension funds to pay pension benefits at all times (column 3a). For example, someone who thinks (s)he is very knowledgeable of financial matters is 3 percentage points more likely to completely trust his/her banks to be able to repay deposits at all times than someone who is financially unknowledgeable.

The financial knowledge effect is robust to the inclusion of generalized trust (columns 1b, 2b and 3b). People who in general trust most other people have more trust in their banks, life insurance company and pension funds than people with low trust; this finding provides support for H3. For example, they are 9 percentage points more likely to have complete trust in the financial health of their banks.

Broad-scope trust is strongly positively related to narrow-scope trust (columns 1c, 2c and 3c). This result implies support for H4a and not for H4b. For example, the likelihood that someone completely trusts the financial soundness of one's own bank(s) is 72% if one also has complete trust in the financial soundness of banks in general, whereas it is 25% if one predominantly trusts banks in general. The regressions that include broad-scope trust also reveal that part of the financial literacy effect takes place indirectly via higher broad-scope trust. The effect of financial knowledge is smaller in these regressions (columns 1c, 2c, and 3c) than in the regressions without broad-scope trust (columns 1b, 2b, and 3b). In case of insurance companies (column 2c) the direct financial knowledge effect is even insignificant.

We do not find support for H11 (financial knowledge has a greater positive effect on narrow-scope trust when broad-scope trust is low than when broad-scope trust is high). For banks and pension funds the coefficient on the interaction term financial knowledge*broad-scope trust in banks is insignificant (columns 1c and 3c). For life insurance companies, it is positive and significant (column 2c). The latter implies that financial knowledge has a greater positive effect on narrow-scope trust when broad-scope trust is high than when broad-scope trust is low.

Narrow-scope trust varies between people. Males have more trust in their financial institutions than females. For example, a male is 3 percentage points more likely to completely trust his own bank(s) (based on column 1c). Compared to people younger than 35, older people are less likely to trust the financial health of their banks and insurance companies. In contrast, trust in one's own pension fund(s) increases with age. For example, someone who is 65 or over is 7 percentage points more likely to completely trust the financial soundness of one's own pension fund(s) than someone who is younger than 35 (based on column 3b). People with a job have less trust in their own life insurance and pension fund than people without a job. Trust in one's own life insurance company is relatively high for people who are living together with a partner or are part of a household where the household head has a partner. However, the presence of a partner is not significantly related to narrow-scope trust in banks and narrow-scope trust in pension

funds. Narrow-scope trust is also unrelated to household income. For all types of narrow-scope trust it does not matter where one lives: the coefficients on region dummies and the degree of urbanization variable are insignificant. When we include the broad-scope trust variables, we find that the stronger people's degree of risk aversion is, the stronger trust in the financial soundness of one's bank, life insurance company and pension fund.

4.4 Financial knowledge is also key for trust in the competence and integrity of managers

Finally, we show that financial knowledge is also positively related to trust in the competence and integrity of managers of financial initiations (Table 6), so H9 is confirmed. Compared to people who find themselves unknowledgeable of financial matters, people who think they are very knowledgeable are 6 percentage points more likely to agree or completely agree that managers are competent (column 2a) and 3 percentage points more likely to agree or completely agree that managers are integer (column 3a).

The effect of financial knowledge is smaller when the average narrow-scope trust measure is included (columns 1b, 2b and 3b). We find support for H5a: there is a positive relationship between narrow-scope trust and trust in the competence and integrity of managers of financial institutions. For example, a one-point higher narrow-scope trust goes along with a 15 percentage points higher likelihood of agreeing that managers are competent and a 2 percentage higher likelihood of completely agreeing.

We also find support for H5b: there is a positive relationship between broad-scope trust and trust in the competence and integrity of managers of financial institutions (columns 1c, 2c, 3c). To illustrate the effect, a one point higher broad-scope trust goes along with a 14 percentage points higher likelihood of agreeing or completely agreeing that managers of financial institutions are integer.

Trust in financial institutions' managers also relates to various sociodemographic factors. In this discussion, we focus on the 2010-2019 findings, when trust in the competence and integrity was measured separately. Males have less trust in financial institutions' managers than females. Men are 4 percentage points less likely to agree or completely agree that these managers are competent (based on column 2a) and 3 percentage points less likely to agree or fully agree that they are integer (based on column 3a) than women. Recall that, in contrast, men had more trust in the financial soundness of financial institutions. Age also matters. Compared to people below 35, people of 45 and above are less likely to find managers of financial institutions competent. Compared to young people, people aged 65 or above have more trust in the integrity of these managers. But this effect is insignificant when either narrow-scope trust or broad-scope trust is included. Compared to low-income people, people in the highest income class are more likely to

	Trust in 1	managers' co	mpetence	Trust in r	nanagers' co	mpetence	Trust in	managers'	integrity
	11 450 111 1	and integrity	7	11 use in 1	nunugers ee	mpetence	11 use in	managers	integrity
	(1a)	(1b)	(1c)	(2a)	(2b)	(2c)	(3a)	(3b)	(3c)
Financial knowledge	0.33***	0.22***	0.20***	0.14***	0.11***	0.09**	0.10**	0.09**	0.07*
	(0.06)	(0.06)	(0.06)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
Male	-0.05	-0.19**	-0.26***	-0.19***	-0.29***	-0.31***	-0.23***	-0.32***	-0.33***
	(0.09)	(0.09)	(0.09)	(0.07)	(0.07)	(0.07)	(0.07)	(0.07)	(0.07)
Between 35 and 44	0.10	0.12	0.00	-0.03	-0.06	-0.12	0.12	0.09	0.05
	(0.16)	(0.15)	(0.15)	(0.11)	(0.10)	(0.10)	(0.10)	(0.10)	(0.10)
Between 45 and 54	0.16	0.11	-0.03	-0.32***	-0.38***	-0.44***	-0.04	-0.12	-0.17
Determine FF and CA	(0.15)	(0.14)	(0.14)	(0.11)	(0.11)	(0.10)	(0.11)	(0.10)	(0.10)
Between 55 and 64	-0.09	-0.21	-0.31**	-0.39^{+++}	-0.59^{+++}	-0.63^{+++}	0.03	-0.18*	-0.20*
65 and over	(0.16)	(0.15)	(0.15)	(0.11)	(0.10)	(0.10)	(U.11) 0 20***	(0.10)	(0.10)
65 una over	-0.17	$-0.54^{\circ\circ}$	-0.28°	-0.03	-0.29^{11}	-0.56	(0.12)	(0.10)	(0.10)
Education: hacholor or higher	0.10)	0.175	0.12*	0.12)	0.06	0.04	(0.12)	(0.11)	(0.11)
Education. Dachelor of higher	(0.20	(0.20)	(0.10)	(0.08)	(0.00)	(0.04)	-0.03	(0.09)	(0.09)
Income: FIIR 1151-1800	-0.01	-0.04	-0.25	0.12	0.00	-0.01	0.11	0.13	0.03
Income. LON 1151-1000	(0.24)	(0.22)	(0.20)	(0.12)	(0.00)	(0.14)	(0.13)	(0.13)	(0.03
Income: FIIR 1801-2600	-0.25	-0.34	-0 52***	0.18	0.07	-0.04	0.19	0.12	0.01
meome. Bon 1001 2000	(0.24)	(0.22)	(0.20)	(0.16)	(0.15)	(0.14)	(0.14)	(0.12)	(0.13)
Income: > EUR 2600	-0.10	-0.19	-0.40*	0.44***	0.34**	0.22	0.32**	0.26*	0.14
	(0.26)	(0.24)	(0.22)	(0.16)	(0.16)	(0.15)	(0.14)	(0.14)	(0.13)
lob	-0.15	-0.09	-0.23**	0.13*	0.19***	0.11	-0.04	-0.01	-0.08
,	(0.12)	(0.12)	(0.12)	(0.08)	(0.07)	(0.07)	(0.08)	(0.07)	(0.07)
Homeowner	0.05	0.04	0.01	0.13	0.10	0.09	0.11	0.11	0.09
	(0.14)	(0.13)	(0.13)	(0.09)	(0.08)	(0.08)	(0.09)	(0.09)	(0.09)
Household head lives with partner	0.24*	0.21	0.24*	-0.27***	-0.25***	-0.22**	-0.12	-0.11	-0.09
-	(0.14)	(0.13)	(0.13)	(0.10)	(0.09)	(0.09)	(0.09)	(0.08)	(0.08)
Degree of urbanization	-0.02	-0.03	-0.06	-0.03	-0.02	-0.02	-0.05	-0.05	-0.05
	(0.04)	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)
Region: north	0.03	-0.11	-0.09	-0.04	-0.06	-0.09	0.06	0.04	0.02
	(0.18)	(0.16)	(0.16)	(0.14)	(0.13)	(0.12)	(0.15)	(0.14)	(0.14)
Region: east	-0.02	-0.01	0.01	0.03	0.06	0.08	0.04	0.09	0.09
	(0.16)	(0.14)	(0.13)	(0.11)	(0.10)	(0.10)	(0.11)	(0.10)	(0.10)
Region: south	0.24*	0.20	0.24*	0.16	0.20*	0.21**	0.03	0.07	0.07
	(0.14)	(0.13)	(0.13)	(0.12)	(0.10)	(0.10)	(0.11)	(0.11)	(0.10)
Risk aversion	-0.09**	-0.11***	-0.09**	-0.08***	-0.10***	-0.08***	-0.07**	-0.08***	-0.07**
W A I I I	(0.04)	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)
Year after bankruptcy		-0.16	-0.30	0.38	0.18	0.13	$0.5/^{**}$	0.42	0.34
Very after hailent	(0.55)	(0.48)	(0.46) 1 5 6 * * *	(0.26)	(0.27)	(0.25)	(0.27)	(0.26)	(0.27)
rear after ballout	-2.21	$-1.02^{+1.0}$	$-1.50^{-1.1}$	-0.80^{+++}	$-0.70^{-0.1}$	-0.73^{+++}	-0.28°	-0.18	-0.21
Narrow scope trust, average	(0.12)	(0.13) 1.60***	(0.15)	(0.10)	(0.10)	(0.17)	(0.14)	(0.15) 1 12***	(0.15)
Nurrow-scope crust. average		(0.07)			(0.04)			(0.04)	
Broad-scope trust: average		[0.07]	2 በ6***		(0.04)	1 5 7***		[0.04]	1 //0***
biouu-scope trust. uveruge			2.00			(0.05)			(0.05)
			[0.00]			[0.05]			(0.05)
Number of observations	5,289	5.276	5.276	17,187	17,123	17,155	17,154	17,092	17,119
Number of respondents	2,173	2,170	2.170	4,599	4,593	4.596	4.577	4,570	4.572
Wald γ^2	409.63***	836.11***	998.31***	116.47***	895.02***	1152.42***	73.64***	877.51***	1002.63***
Period	2006-2000	2006-2009	2006-2009	2010-2019	2010-2019	2010-2019	2010-2019	2010-2010	2010-2019

Table 6. Financial knowledge and trust in financial sector managers' competence and integrity

Note: The table reports parameter estimates of random effects ordered logit regressions. Standard errors are clustered by household and shown in parentheses. The dependent variables range from 1 (completely disagree that managers of financial institutions are in general knowledgeable and/or sound) to 5 (completely agree). ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively.

find managers of financial institutions competent and integer. Trust in the competence is also relatively high for people with a paid job and people who live in the south of the Netherlands. The effect of education is insignificant and the effect of having a partner is negative. We also find that stronger risk aversion goes along with lower trust in managers' competence and integrity. Trust in managers is rather low for people who experienced a bailout of their bank in the prior year but high for people who experienced a bankruptcy of their bank. However, the latter effect is not robust to the inclusion of either narrow-scope trust or broad-scope trust.

4.5 Robustness

As a robustness test we rerun the regressions with alternative financial knowledge variables. First, we use *manager household finances* instead of the self-assessed financial knowledge variable. Manager household finances is a binary dummy that is 1 for household members who are most involved in household finances and 0 for other household members. Second, we use *financial sector job* instead of financial knowledge. Financial sector job is 1 for people working in the financial sector and 0 for people without such a job. The results of these analyses are available upon request.

Regarding the first robustness test, when we include manager household finances instead of financial knowledge we also find support for H7, H8, H9 and H10. Compared to household members who are not in charge of household finances, people who are most involved in household finances have a higher degree of broad-scope trust, narrow-scope trust, trust in the competence and integrity of financial sector managers and trust in the supervisor. Our other findings are also the same as before. So, we find support for H3, H4a, H5a, H5b and H6 and no support for H11.

Our baseline results are also confirmed when we use financial sector job instead. Compared to other people, people with a financial sector job have a higher degree of broad-scope trust, narrow-scope trust, trust in the competence and integrity of managers of financial institutions and in DNB.

As another robustness test, we use more general broad-scope measures of trust in the financial sector in general, which do not focus on the financial health of financial institutions. These measures are only available for 2019. *Trust in the financial sector* is an ordered variable capturing trust in the health of financial institutions such as banks, insurance companies and pension funds. It ranges from 1 (absolutely no trust) to 4 (a lot of trust) and is on average 2.4 for the observations included in the regressions. The other trust measure *trust in the financial sector* 2 is a binary dummy. It is 1 for respondent who think that in general most financial institutions such as banks, insurance companies and pension funds can be trusted and 0 for respondents who believe that one cannot be careful enough in dealing with financial institutions. On average it is 0.44 for the respondents included in the regressions, so 44% of the people think that in general financial institutions can be trusted.

Our findings are robust to the use of these alternative measures. Again, we find that better financial knowledge and more trust in the supervisor go along with a significantly higher degree

of broad-scope trust (Appendix B, Table B.3). And, also in line with our previous findings, narrowscope trust in the various types of financial institutions and trust in financial sector managers both positively depend on broad-scope trust. The latter results are available upon request.

5. Conclusions and policy implications

To sum up, using longitudinal trust data this research shows that financial literacy is key for trust in financial institutions. Compared to financially illiterate consumers, consumers with a high degree of financial knowledge are more likely to trust banks, insurance companies and pension funds, and the competence and integrity of the managers of these institutions. This holds both for general trust (broad-scope trust) and trust in one's own financial institutions (narrow-scope trust), which is significantly higher. Financially knowledgeable people are also more likely to trust the prudential supervisor. This finding is robust to the usage of three different financial literacy measures: self-assessed financial knowledge, being in charge of household finances and working in the financial sector.

We also find that people with a higher degree of trust in the financial health of financial institutions in general are also more likely to trust the financial health of their own banks, life insurance company and pension funds. In other words, there is a positive link between broad-scope and narrow-scope trust. For all types of institutions, we also find a positive relationship between generalized trust (trust in other people) and narrow-scope trust.

In addition, people with high narrow-scope or broad-scope trust are more likely to trust the integrity and competence of managers of financial institutions than people with low trust in financial institutions. Furthermore, we find that higher levels of trust in the banking supervisor go along with higher levels of broad-scope trust.

Last, our findings provide valuable insights into the role of consumers' background characteristics. Several of these relationships depend on the type of trust that is researched. For example, compared to females, males are relatively likely to trust the financial health of financial institutions but unlikely to trust the competence and integrity of these institutions.

For central banks and other banking supervisors, this research highlights two ways to strengthen trust in the financial sector. The first way is to support or even develop policies that contribute to financial literacy. The second approach is to act in a way that not only supports trust in financial institutions directly but also indirectly via higher trust in the supervisor itself.

References

- Aghion, P., Y. Algan, P. Cahuc and A. Shleifer (2010). Regulation and distrust. *Quarterly Journal of Economics*, 125(3), 1015–49.
- Agnew, J., L. Szykman, L., S.P. Utkus and J.A. Young (2007). Do financial literacy and mistrust affect 401(k) participation? Center for Retirement Research Issue Brief 7–17. Available at: <u>http://crr.bc.edu/category/briefs/</u>.
- Ampudia, M. and S. Palligkinis (2018). Trust and the household-bank relationship. ECB Working Paper Series No. 2184.
- Balloch, A., A. Nicolae and D. Philip (2015). Stock market literacy, trust, and participation. *Review of Finance*, 19, 1925–1963.
- Brown, M., B. Guin and S. Morkoetter (2017). Deposit withdrawals from distressed commercial banks: Client relationships matter. University of St. Gallen, School of Finance Research Paper No. 2013/19. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2354197
- Brunetti, M., R. Ciciretti and Lj. Djordjevic (2016). The determinants of household's bank switching. *Journal of Financial Stability*, 26, 175–189.
- Butler, J., P. Giuliano and L. Guiso (2016). The right amount of trust. *Journal of the European Economic Association*, 14(5), 1155–1180.
- Carbó-Valverde, S., E. Maqui-Lópe and F. Rodriguez-Fernandez (2013). Trust in banks: evidence from the Spanish financial crisis. 26th Australasian Finance and Banking Conference 2013.
- Deuflhard, F., D. Georgarakos and R. Inderst (2019). Financial literacy and savings account returns. *Journal of the European Economic Association*, 17(1), 131–164.
- Dick, A.S. and K. Basu (1994). Customer loyalty: toward an integrated conceptual framework. *Journal of the Academy of Marketing Science*, 22(2), 99–113.
- Diepstraten, M. and C. van der Cruijsen (2019). To stay or go? Consumer bank switching behaviour after government interventions. *Journal of Banking and Finance*, 106, 16–33.
- Fungáčová, Z., I. Hasan and L. Weill (2019). Trust in banks. *Journal of Economic Behavior & Organization*, 157, 452–476.
- Guiso, L., P. Sapienza and L. Zingales (2006). Does culture affect economic outcomes? *Journal of Economic Perspectives*, 20(2), 23–48.
- Guiso, L. (2010). A trust-driven financial crisis. Implications for the future of financial markets. Einaudi Institute for Economic and Finance, Working Paper No. 1006. Available at: <u>http://ideas.repec.org/p/eie/wpaper/1006.html</u>.
- Hansen, T. (2012). Understanding trust in financial services: the influence of financial healthiness, knowledge, and satisfaction. *Journal of Service Research*, 15(3), 280–295.
- Inglehart, R., C. Haerpfer, A. Moreno, C. Welzel, K. Kizilova, J. Diez-Medrano, M. Lagos, P. Norris, E. Ponarin and B. Puranen et al. (eds.). 2014 World Values Survey: Round Six Country-Pooled

Datafile Version: <u>http://www.worldvaluessurvey.org/WVSDocumentationWV6.jsp</u>. Madrid: JD Systems Institute.

- Jaffer, S., N. Morris and D. Vines (2014). Why trustworthiness is important. In: Capital Failure: Rebuilding Trust in Financial Services, Morris, N. and D. Vines (eds.), Oxford, Oxford University Press.
- Jansen, D-J., R. Mosch and C. van der Cruijsen (2015). When does the general public lose trust in banks? *Journal of Financial Services Research*, 48(2), 127–141.
- Jiang, D. and S.S. Lim (2018). Trust and household debt. *Review of Finance*, 22(2), 783–812.
- Knell, M. and H. Stix (2015). Trust in banks during normal and crisis times: evidence from survey data. *Economica*, 82, 995–1020.
- Lusardi, A. and O.S. Mitchell (2014). The economic importance of financial literacy: theory and evidence. *Journal of Economic Literature*, 52(1), 5–44.
- Mosch, R.H.J. and H.M. Prast (2008). Confidence and trust: empirical investigations for the Netherlands and the financial sector. DNB Occasional Study 6(2). Available at: https://www.dnb.nl/binaries/OS Vol6 2 08 tcm46-175279.pdf
- Pirson, M. and D.K. Malhotra (2008). Unconventional insights for managing stakeholder trust. *MIT Sloan Management Review*, 49(4), 43–50.
- Sapienza, P. and L. Zingales (2012). A trust crisis. *International Review of Finance*, 12(2), 123–131.
- Sapienza, P. and L. Zingales (2018). Financial trust index. Available at: <u>http://www.financialtrustindex.org/</u>.
- Sirdeshmukh, D., J. Singh and B. Sabol (2002). Consumer trust, value, and loyalty in relational exchanges. *Journal of Marketing*, 66(1), 15–37.
- Stevenson, B. and J. Wolfers (2011). Trust in public institutions over the business cycle. *American Economic Review: Papers & Proceedings*, 101(3), 281–287.
- Stix, H. (2013). Why do people save in cash? Distrust, memories of banking crises, weak institutions and dollarization, *Journal of Banking and Finance*, 37(1), 4087–4106.
- Teppa, F. and C. Vis (2012). The CentERpanel and the DNB Household Survey: Methodologicalaspects.DNBOccasionalStudyNo.10(4).Availableat:http://www.dnb.nl/binaries/DNB_OS_1004_BIN_WEB_tcm46-277691.pdf
- van der Cruijsen, C. (2019). Payments data: do consumers want banks to keep them in a safe or turn them into gold? *Applied Economics*, https://doi.org/10.1080/00036846.2019.1659493.
- van der Cruijsen, C., J. de Haan and D-J. Jansen (2016). Trust and financial crisis experiences. *Social Indicators Research*, 127(2), 577–600.
- van der Cruijsen, C. and M. Diepstraten (2017). Banking products: You can take them with you, so why don't you? *Journal of Financial Services Research*, 52(1), 123–154.

- van der Cruijsen, C., M. Doll and F. van Hoenselaar (2019). Trust in other people and the usage of peer platform markets. *Journal of Economic Behavior & Organization*, 166, 751–766.
- van der Cruijsen, C. and N. Jonker (2019). Pension profile preferences: the influence of trust and expected expenses. *Applied Economics*, 51(12), 1212–1231.
- van Dalen, H.P. and K. Henkens (2018). The making and breaking of trust in pension providers: an empirical study of pension participants. *Geneva Papers on Risk and Insurance*, 43(3), 473–491.
- van Esterik-Plasmeijer, P.W.J. and W.F. van Raaij (2017). Banking system trust, bank trust, and bank loyalty. *International Journal of Bank Marketing*, 35(1), 97–111.
- van Rooij, M., A. Lusardi and R. Alessie (2011). Financial literacy and stock market participation. *Journal of Financial Economics*, 101(2), 449–472.
- Von Gaudecker, H-M. (2015). How does household portfolio diversification vary with financial literacy and financial advice? *The Journal of Finance*, 70(2), 489–507.

Appendix A. Description of variables

Table A.1 Description of variables included in the baseline regression analyses (1/2)

Variable	Description	Mean	Sd	Min	Max	N
Key variables						
Broad-scope trust in banks	Answer to 'In general, do you trust that banks in the Netherlands are able to repay deposits at all times?' (1 =					
	no, not at all, 2 = no, predominantly not, 3 = neutral, 4 = yes, predominantly, 5 = yes, completely).	3.72	0.82	1	5	23,047
Broad-scope trust in insurance	Answer to 'In general, do you trust that insurance companies in the Netherlands are able to fulfil their					
companies	payment obligations to all persons insured at all times?' (1 = no, not at all, 2 = no, predominantly not, 3 =					
	neutral, 4 = yes, predominantly, 5 = yes, completely).	3.66	0.78	1	5	21,949
Broad-scope trust in pension funds	Answer to the question 'In general, do you trust pension funds in the Netherlands to fulfil their payment					
	obligations towards retirees at all times?' (1 = no, not at all, 2 = no, predominantly not, 3 = neutral, 4 = yes,					
_	predominantly, 5 = yes, completely.)	3.42	0.91	1	5	22,761
Trust in the supervisor	Ordered variable capturing trust in DNB (1 = absolutely no trust, 2 = not so much trust, 3 = pretty much trust,					
	4 = a lot of trust).	2.71	0.75	1	4	23,719
Narrow-scope trust in banks	Answer to 'At the moment, do you trust that the bank(s) at which you have deposits is (are) able to repay					
	these deposits at all times?' (1 = no, not at all, 2 = no, predominantly not, 3 = neutral, 4 = yes, predominantly,			_	_	
	5 = yes, completely).	3.98	0.83	1	5	23,181
Narrow-scope trust in own life	Answer to 'At the moment, do you trust that the life insurance company at which you have contracts is able to					
insurance company	pay your insurance money at all times? $(1 = no, not at all, 2 = no, predominantly not, 3 = neutral, 4 = yes,$	0.04	0 = 0		_	0.04.0
N7	predominantly, 5 = yes, completely).	3.94	0.79	1	5	9,813
Narrow-scope trust in own pension	Answer to the question Do you trust your pension fund(s) to be able to pay your pension benefit at all	0.70	0.04		_	4 - (00
funds	times? (1 = no, not at all, 2 = no, predominantly not, 3 = neutral, 4 = yes, predominantly, 5 = yes, completely.)	3.70	0.94	1	5	17,630
I rust in managers' competence	Agreement with Managers of financial institutions are in general knowledgeable and sound. (1 = completely	2.24	0.00	1	-	F 200
ana integrity	disagree, 2 = disagree, 3 = neutral, 4 = agree, 5 = completely agree).	3.34	0.80	1	5	5,289
Trust in managers competence	Agreement with Managers of financial institutions are in general knowledgeable. $(1 = \text{completely disagree}, 2)$	2.20	0.02	1	-	17 107
Truct in managane' integrity	= clisagree, 3 = neutral, 4 = agree, 5 = completely agree).	3.20	0.83	1	5	17,187
Trust in managers integrity	Agreement with Managers of manifold institutions are in general sound. $(1 = \text{completely usagive}, 2 = diagram 2 = noticely as a completely $	2 0 2	0.07	1	F	17154
Droad coope truct, querage	u_{13} and u_{13} and u_{14} and u_{14} and u_{13} and u_{14} and $u_{$	2.05	0.07	1	Э	17,154
Brouu-scope trust: uveruge	The average of the unreferred bload-scope if ust measures. For each respondent this average is based on the	2.60	060	1	E	22 670
Narrow scope trust, guarage	available bload-scope if usi measures, so at most mee.	5.00	0.00	1	5	22,070
Nurrow-scope trust. uveruge	The average of the unreferrent harrow-scope if ust measures. For each respondent this average is based on the	2.00	0.75	1	5	22 645
Financial Imouladae	Solf accessed knowledge of financial matters (1 = not knowledgeable 2 = more or less knowledgeable 2 =	5.00	0.75	1	5	22,045
rinunciui knowieuge	self-assessed knowledge of financial matters (1 – not knowledgeable, 2 – more of financial matters is not					
	$\frac{1}{2}$ available yet. Therefore, we take the most recent available data, accuming that the colf accossment did not					
	avanable yet. Therefore, we take the most recent available tata, assuming that the self-dssessifient the not	214	0.73	1	1.	23 722
	change.	2.14	0.75	T	ч	43,144

Note: This table describes the variables used in the regressions of which the results are reported in Tables 3, 4, 5 and 6. The mean, standard deviation (sd), minimum (min), maximum (max), and number of observations (N) are reported for the sample included in these regressions.

Variable	Description	Mean	Sd	Min	Max	Ν
<u>Control variables</u>						
Male	Binary dummy (1 = male, 0 = female).	0.55	0.50	0	1	23,722
34 and below	Binary dummy (1 = 34 and below, 0 = else). Reference category.	0.13	0.34	0	1	23,722
Between 35 and 44	Binary dummy (1 = between 35 and 44, 0 = else).	0.16	0.37	0	1	23,722
Between 45 and 54	Binary dummy (1 = between 45 and 54, 0 = else).	0.18	0.38	0	1	23,722
Between 55 and 64	Binary dummy (1 = between 55 and 64, 0 = else).	0.22	0.41	0	1	23,722
65 and over	Binary dummy (1 = 65 and over, 0 = else).	0.31	0.46	0	1	23,722
Education: bachelor or higher	Binary dummy (1 = higher vocational education or university education, 0 = else).	0.38	0.49	0	1	23,722
Income: ≤ EUR 1150	Binary dummy (1 = household net monthly income ≤ EUR 1150, 0 = else). Reference category.	0.05	0.22	0	1	23,722
Income: EUR 1151-1800	Binary dummy (1 = household net monthly income ≥ EUR 1151 and ≤ EUR 1800, 0 = else).	0.17	0.37	0	1	23,722
Income: EUR 1801-2600	Binary dummy (1 = household net monthly income ≥ EUR 1801 and ≤ EUR 2600, 0 = else).	0.29	0.45	0	1	23,722
Income: > EUR 2600	Binary dummy (1 = household net monthly income > EUR 2600, 0 = else).	0.49	0.50	0	1	23,722
Job	Binary dummy (1 = paid job, work in family business or self-employed, 0 = else)	0.51	0.50	0	1	23,722
Homeowner	Binary dummy (1 = homeowner, 0 = else).	0.75	0.43	0	1	23,722
Household head lives with partner	Binary dummy (1 = head of household is living together or married, 0 = else).	0.76	0.43	0	1	23,722
Degree of urbanization	Degree of urbanization of respondent's residence based on the address density per km ² (1 = 500 or less, 2 =					
	500-1000, 3 = 1000-1500, 4 = 1500-2500, 5 = more than 2500).	2.98	1.32	1	5	23,722
Region: west	Binary dummy (1 = living in the west of the Netherlands, 0 = else). Reference category.	0.43	0.49	0	1	23,722
Region: north	Binary dummy (1 = living in the north of the Netherlands, 0 = else).	0.13	0.33	0	1	23,722
Region: east	Binary dummy (1 = living in the east of the Netherlands, 0 = else).	0.21	0.40	0	1	23,722
Region: south	Binary dummy (1 = living in the south of the Netherlands, 0 = else).	0.24	0.43	0	1	23,722
Risk aversion	Average agreement measured on a 1 (totally disagree) to 7 (totally agree) scale with six statements on risk					
	taking: 1) 'I think it is more important to have safe investments and guaranteed returns, than to take a risk to					
	have a chance to get the highest possible returns.', 2) 'I do not invest in shares, because I find this too risky.',					
	3) 'If I think an investment will be profitable, I am prepared to borrow money to make this investment.'					
	(reversed scale), 4) 'I want to be certain that my investments are safe.', 5) 'If I want to improve my financial					
	position, I should take financial risks. (reversed scale), and 6) I am prepared to take the risk to lose money,	F 0 F	1.00	1	-	22 722
Very often han hunnter	when there is also a chance to gain money. (reversed scale).	5.27	1.02	1	/	23,722
Year after bankrupicy	binary dummy that is 1 in the year after a bankruptcy for customers who experienced that their bank went handrupt in the prior year and 0 also	0.01	0.00	0	1	22 E 1 1
Vear after hailout	Binary dummy that is 1 in the year after a hailout for customers who experienced that their hank was hailed	0.01	0.00	0	1	23,344
i cai aj ci bullout	out in the prior year and 0 else.	0.03	0.18	0	1	23.544
Generalized trust	Binary dummy (1 = in general most other people can be trusted, 0 = one cannot be careful enough in dealing	0.00	0110	Ũ	*	20,011
	with people).	0.67	0.47	0	1	23,467

Table A.1 Description of variables included in the baseline regression analyses (2/2)

Note: This table describes the variables used in the regressions of which the results are reported in Tables 3, 4, 5 and 6. The mean, standard deviation (sd), minimum (min), maximum (max), and number of observations (N) are reported for the sample included in these regressions.

Variable	Description	Mean	Sd	Min	Max	Ν
Manager household finances	Binary dummy (1 = household member who is most involved in finances, 0 = else).	0.68	0.46	0	1	23,901
Financial sector job	Binary dummy (1 = working in the financial sector, 0 = no job or a job in another sector).	0.03	0.18	0	1	23,914
Trust in the financial sector	Ordered variable capturing trust in financial institutions such as banks, insurance companies and pension					
	funds (1 = absolutely no trust, 2 = not so much trust, 3 = pretty much trust, 4 = a lot of trust).	2.38	0.69	1	4	1,777
Trust in the financial sector 2	Binary dummy (1 = in general most financial institutions such as banks, insurance companies and pension					
	funds can be trusted, 0 = one cannot be careful enough in dealing with financial institutions).	0.44	0.50	0	1	1,779

Table A.2 Description of variables included in the robustness analyses

Note: This table describes the variables used in the robustness analyses. The mean, standard deviation (sd), minimum (min), maximum (max), and number of observations (N) are reported for the sample included in these regressions.

Appendix B: Additional results

Table B.1. Difference between generalized trust and narrow-scope trust: annual data

															Total
	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	sample
Banks															
(a) Generalized trust 2	3.41	3.71	3.78	3.84	3.61	3.46	3.83	3.77	3.73	3.60	3.54	3.57	3.54	3.55	3.64
(b) Narrow-scope trust	4.41	4.31	4.20	3.89	3.97	3.92	3.91	3.82	3.82	3.77	3.80	3.91	3.87	3.91	3.95
(a) - (b)	-1.00***	-0.60***	-0.42***	-0.06*	-0.36***	-0.46***	-0.08**	-0.04	-0.09***	-0.16***	-0.26***	-0.33***	-0.33***	-0.36***	-0.32***
Number of observations	1,931	2,046	1,918	1,882	2,067	2,096	2,231	2,025	2,630	2,549	2,073	2,509	2,136	2,567	30,660
Life insurance companies															
(c) Generalized trust 2	3.44	3.75	3.76	3.89	3.68	3.55	4.00	3.88	3.82	3.65	3.58	3.65	3.56	3.49	3.67
(d) Narrow-scope trust	4.31	4.28	4.20	3.82	3.92	3.83	3.83	3.81	3.92	3.97	3.98	4.05	3.71	3.67	3.92
(c) - (d)	-0.88***	-0.53***	-0.45***	0.07	-0.24***	-0.27***	0.12**	0.06	-0.10**	-0.32***	-0.40***	-0.40***	-0.15***	-0.18***	-0.25***
Number of observations	841	869	845	751	818	823	845	730	1,081	898	672	802	1,441	1,803	13,219
Pension funds															
(e) Generalized trust 2	3.50	3.79	3.87	3.93	3.74	3.58	3.93	3.86	3.84	3.70	3.64	3.67	3.64	3.68	3.74
(f) Narrow-scope trust	4.03	4.12	4.17	3.63	3.79	3.59	3.42	3.44	3.49	3.55	3.59	3.60	3.68	3.67	3.68
(e) - (f)	-0.53***	-0.32***	-0.30***	-0.30***	-0.04	-0.02	0.51***	0.42***	0.36***	0.15***	0.04	0.07*	-0.04	0.01	0.06***
Number of observations	1,303	1,380	1,345	1,325	1,517	1,547	1,702	1,529	2,017	1,842	1,587	1,876	1,559	1,853	22,382

Note: ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively (one-sided paired t-test).

							0010						0010		Total
	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	sample
Banks															
(a) Narrow-scope trust	4.41	4.31	4.20	3.90	3.98	3.93	3.91	3.82	3.82	3.77	3.81	3.91	3.87	3.91	3.96
(b) Broad-scope trust	4.12	4.04	3.98	3.58	3.65	3.58	3.60	3.54	3.56	3.57	3.59	3.68	3.66	3.70	3.70
(a) - (b)	0.29***	0.27***	0.22***	0.32***	0.32***	0.35***	0.31***	0.27***	0.26***	0.20***	0.22***	0.23***	0.22***	0.21***	0.26***
Number of observations	1,911	2,032	1,908	1,863	2,047	2,071	2,205	2.002	2,610	2,531	2,045	2,479	2,111	2,543	30,358
(Life) insurance companies															
(c) Narrow-scope trust	4.32	4.28	4.21	3.83	3.93	3.82	3.84	3.82	3.92	3.98	3.98	4.06	3.71	3.68	3.93
(d) Broad-scope trust	4.06	4	3.98	3.63	3.74	3.64	3.60	3.64	3.74	3.76	3.76	3.85	3.68	3.66	3.76
(c) - (d)	0.26***	0.28***	0.23***	0.20***	0.18***	0.18***	0.24***	0.18***	0.19***	0.21***	0.22***	0.21***	0.03**	0.02*	0.17***
Number of observations	835	866	839	739	814	817	837	721	1,072	884	671	789	1,429	1,780	13,093
Donaion funda															
Pension lunus															
(e) Narrow-scope trust	4.04	4.12	4.17	3.63	3.79	3.59	3.42	3.43	3.49	3.55	3.60	3.60	3.68	3.67	3.68
(f) Broad-scope trust	3.85	3.91	3.95	3.37	3.57	3.34	3.13	3.20	3.28	3.38	3.38	3.43	3.47	3.46	3.46
(e) - (f)	0.18***	0.21***	0.22***	0.25***	0.22***	0.26***	0.29***	0.24***	0.20***	0.17***	0.22***	0.17***	0.21***	0.22***	0.22***
Number of observations	1,286	1,367	1,343	1,315	1,502	1,537	1,694	1,518	2,001	1,827	1,575	1,863	1,546	1,839	22,213

Table B.2. Narrow-scope and broad-scope trust: annual data

Note: ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively (one-sided paired t-test).

	Trust in the fi	nancial sector	Trust in the fir	nancial sector 2
	(1a)	(1b)	(2a)	(2b)
Financial knowledge	0.26***	0.13*	0.27***	0.19**
0	(0.07)	(0.07)	(0.07)	(0.08)
Male	-0.25***	-0.52***	0.03	-0.17
	(0.09)	(0.10)	(0.10)	(0.11)
Between 35 and 44	-0.01	-0.06	0.25	0.27
	(0.18)	(0.19)	(0.20)	(0.21)
Between 45 and 54	-0.29*	-0.34*	-0.12	-0.16
	(0.17)	(0.18)	(0.18)	(0.20)
Between 55 and 64	-0.35**	-0.40**	-0.02	0.01
	(0.18)	(0.19)	(0.19)	(0.20)
65 and over	-0.03	-0.25	0.14	0.03
	(0.19)	(0.20)	(0.20)	(0.21)
Education: bachelor or higher	0.05	-0.22*	0.27**	0.10
	(0.10)	(0.11)	(0.11)	(0.12)
Income: EUR 1151-1800	-0.22	-0.34	0.07	0.11
	(0.24)	(0.25)	(0.24)	(0.26)
Income: EUR 1801-2600	-0.09	-0.21	0.29	0.33
	(0.23)	(0.24)	(0.23)	(0.25)
Income: > EUR 2600	0.06	-0.17	0.49**	0.44*
	(0.25)	(0.25)	(0.24)	(0.26)
Job	0.17	0.13	0.14	0.12
	(0.14)	(0.14)	(0.14)	(0.16)
Homeowner	0.31**	0.18	0.26**	0.15
	(0.13)	(0.13)	(0.13)	(0.14)
Household head lives with partner	-0.06	0.07	-0.28**	-0.23
	(0.13)	(0.13)	(0.13)	(0.14)
Degree of urbanization	-0.00	-0.02	-0.01	-0.03
	(0.04)	(0.04)	(0.04)	(0.05)
Region: north	-0.10	-0.10	0.06	0.11
	(0.15)	(0.16)	(0.18)	(0.19)
Region: east	-0.03	0.02	0.09	0.16
	(0.14)	(0.14)	(0.14)	(0.15)
Region: south	0.03	0.16	0.07	0.19
	(0.13)	(0.13)	(0.13)	(0.14)
Risk aversion	-0.04	-0.03	-0.01	-0.01
	(0.05)	(0.05)	(0.05)	(0.06)
Trust in the supervisor		2.11***		1.71***
		(0.11)		(0.12)
Constant			-1.36***	-5.89***
			(0.44)	(0.58)
Number of observations	1,777	1,777	1,779	1,777
Wald χ^2	42.90***	423.28***	57.39***	237.28***
Period	2019	2019	2019	2019
Model	Ordered logit	Ordered logit	Logit	Logit

Table B.3 Alternative measures of	broad-scope trust	(2019 only)

Note: The table reports parameter estimates. Standard errors are clustered by household and shown in parentheses. The dependent variables of models (1) and (2) range from 1 (absolutely no trust) to 4 (a lot of trust). The dependent variable of models (3) and (4) are binary dummies (1 = trust, 0 = no trust). ***, ** and * denote statistical significance at the 0.01, 0.05 and 0.10 level respectively.

Previous DNB Working Papers in 2019

No. 622	David-Jan Jansen, Did Spillovers From Europe Indeed Contribute to the 2010 U.S. Flash
No (a)	Willing Balt Kastas Marromatis and Swadar wan Wiinharson The Clobal Macrosson amice
100. 623	of a trade war: the EAGLE model on the US-China trade conflict
No. 624	Ronald Heijmans and Chen Zhou, Outlier detection in TARGET2 risk indicators
No. 625	Robert Vermeulen, Edo Schets, Melanie Lohuis, Barbara Kölbl, David-Jan Jansen
	and Willem Heeringa, The Heat is on: A framework measuring financial stress under
	disruptive energy transition scenarios
No. 626	Anna Samarina and Anh D.M. Nouven. Does monetary policy affect income inequality in
1101 020	the euro area?
No 627	Stanhania Titzak and Ian Willow van dan End. The impact of size composition and
100.02/	duration of the control hand helenes short on inflation expositions and market prices
NI - (-0	a 1 o 1 · · · · · · · · · · · · · · · · ·
INO. 628	Andrea Colciago, Volker Lindenthal and Antonella Irigari, who Creates and Destroys
	Jobs over the Business Cycle?
No. 629	Stan Olijslagers, Annelie Petersen, Nander de Vette and Sweder van Wijnbergen, What
	option prices tell us about the ECB's unconventional monetary policies
No. 630	Ilja Boelaars and Dirk Broeders, Fair pensions
No. 631	Joost Bats and Tom Hudepohl, Impact of targeted credit easing by the ECB: bank-
	level evidence
No. 632	Mehdi El Herradi and Aurélien Leroy, Monetary policy and the top one percent:
	Evidence from a century of modern economic history
No. 633	Arina Wischnewsky, David-Jan Jansen and Matthias Neuenkirch, Financial
55	Stability and the Fed: Evidence from Congressional Hearings
No. 634	Bram Gootjes, Jakob de Haan and Richard Jong-A-Pin, Do fiscal rules constrain
51	political budget cycles?
No. 635	Jasper de Jong and Emmanuel de Veirman. Heterogeneity and Asymmetric
_ (_ (_))	Macroeconomic effects of changes in Loan-to-Value limits
No. 636	Niels Gilbert , Euro area sovereign risk spillovers before and after the ECB's OMT
No. 627	Dorinth van Diik Local Constant-Quality Housing Market Liquidity Indices
No. 628	Francesco G. Caolia Mauro Mastrogiacomo and Giacomo Pasini Being in Good
110.050	Hands: Deposit Insurance and Peers Financial Sophistication
No 620	Maurice Bun and Jasner de Winter Measuring trends and persistence in capital and
100.039	labor missilogation
No (co	Elorion Unice Michael Hurd Maarton van Daaii Tahias Desemann and Ioashim
110. 640	Winter Dynamics and betere consists of subjective steals market superstations
N.	Densions Chan and Reading on With homen Dedictive stock market expectations
INO. 641	Damiaan Unen and Sweder van Wijnbergen, Redistributive Consequences of
	Abolishing Uniform Contribution Policies in Pension Funds
No. 642	Richard Heuver and Ron Triepels, Liquidity Stress Detection in the European Banking Sector
No. 643	Dennis Vink, Mike Nawas and Vivian van Breemen, Security design and credit rating
	risk in the CLO market
No. 644	Jeroen Hessel, Medium-term Asymmetric Fluctuations and EMU as an Optimum Currency Area
No. 645	Dimitris Christelis, Dimitris Georgarakos, Tullio Jappelli, Luigi Pistaferri and
	Maarten van Rooij, Wealth Shocks and MPC Heterogeneity
No. 646	Dirk Bezemer and Anna Samarina, Debt Shift, Financial Development and Income Inequality
No. 647	Jan Willem van den End, Effects of QE on sovereign bond spreads through the safe asset channel
No. 648	Bahar Öztürk and Ad Stokman. Animal spirits and household spending in Europe and the US
· · · · · · · · · · · · · · · · · · ·	

- No. 649 Garyn Tan, Beyond the zero lower bound: negative policy rates and bank lending
- No. 650 Yakov Ben-Haim and Jan Willem van den End, Fundamental uncertainty about the natural rate of interest: Info-gap as guide for monetary policy
- No. 651 Olivier Coibion, Dimitris Georgarakos, Yuriy Gorodnichenko and Maarten van Rooij, How does consumption respond to news about inflation? Field evidence from a randomized control trial
- No. 652 Nikos Apokoritis, Gabriele Galati, Richhild Moessner and Federica Teppa, Inflation expectations anchoring: new insights from micro evidence of a survey at high-frequency and of distributions
- No. 653 **Dimitris Mokas and Rob Nijskens,** Credit risk in commercial real estate bank loans: the role of idiosyncratic versus macro-economic factors
- No. 654 **Cars Hommes, Kostas Mavromatis, Tolga Özden and Mei Zhu,** Behavioral Learning Equilibria in the New Keynesian Model
- No. 655 **Leo de Haan and Mauro Mastrogiacomo,** Loan to value caps and government-backed mortgage insurance: loan-level evidence from Dutch residential mortgages
- No. 656 Jakob de Haan, Zhenghao Jin and Chen Zhou, Micro-prudential regulation and banks' systemic risk
- No. 657 Michael Kurz and Stefanie Kleimeier, Credit Supply: are there negative spillovers from banks' proprietary trading?
- No. 658 Patty Duijm, Foreign funded credit: funding the credit cycle?
- No. 659 Mary Everett, Jakob de Haan, David-Jan Jansen, Peter McQuade, and Anna Samarina, Mortgage lending, monetary policy, and prudential measures in small euro-area economies: Evidence from Ireland and the Netherlands
- No. 660 **Tom Hudepohl, Ryan van Lamoen and Nander de Vette,** Quantitative Easing and Exuberance in Stock Markets: Evidence from the euro area
- No. 661 Paul Pelzl and María Teresa Valderrama, Capital Regulations and the Management of Credit Commitments during Crisis Times

DeNederlandscheBank

De Nederlandsche Bank N.V. Postbus 98, 1000 AB Amsterdam 020 524 91 11 dnb.nl